Author’s response to reviews

Title: High prevalence of diarrheagenic Escherichia coli carrying toxin-encoding genes isolated from children and adults in southeastern Brazil

Authors:

Liliana Spano (liliana.spano@ufes.br)
Keyla Cunha (fonseca.keyla@gmail.com)
Mariane Monfardini (marianemonfardini@yahoo.com.br)
Rita de Cássia Fonseca (fonseca.vix@terra.com.br)
Isabel Christina Scaletsky (scaletsky@unifesp.br)

Version: 2 Date: 29 Sep 2017

Author’s response to reviews:

Dear Dr Benny Yau

BMC Infectious Diseases

We are sending the manuscript entitled “High prevalence of diarrheagenic Escherichia coli carrying toxin-encoding genes isolated from children and adults in southeastern Brazil” adjusted as requested.

Yours sincerely,

Liliana Cruz Spano